Author's response to reviews

Title: The effect of gold kiwifruit consumed with an iron fortified breakfast cereal meal on iron status in women with low iron stores: A 16 week randomised controlled intervention study

Authors:

Kathryn Beck (K.L.Beck@massey.ac.nz)
Cathryn Conlon (C.Conlon@massey.ac.nz)
Rozanne Kruger (R.Kruger@massey.ac.nz)
Jane Coad (J.Coad@massey.ac.nz)
Welma Stonehouse (W.Stonehouse@massey.ac.nz)

Version: 5 Date: 22 January 2010

Author's response to reviews:

NA